Author's response to reviews

Title: What ICU nurses in different Austrian hospitals know and think about the Austrian organ donation law

Authors:

Gabriele Zettel (gabriele.zettel@medunigraz.at)
Angela Horvath (angela.horvath@medunigraz.at)
Ekaterina Vorobyeva (ekatarina.vorobyeva@stud.medunigraz.at)
Christian Auburger (c.auburger@hotmail.com)
Michael Zink (michael.zink@bbstveit.at)
Philipp Stiegler (philipp.stiegler@medunigraz.at)
Vanessa Stadlbauer (vanessa.stadlbauer@medunigraz.at)

Version: 4 Date: 9 June 2014

Author's response to reviews:

Point-by-point reply

What ICU nurses in different Austrian hospitals know and think about the Austrian organ donation law

We would like to thank both reviewers for their valuable comments. We have made all changes as suggested. The changes are marked in yellow in the manuscript.

Reviewer 1

1) The number of respondents from both groups is stated in the abstract
2) More details about the questions have been added to the main text.
3) Citations: We are sorry but there is no translation to English available for the Austrian Hospital Act and for the survey conducted by the ministry of health in lower Austria. But we can remove these references if the reviewer and the editor agree that this would be helpful.

Reviewer 2

We have included the suggestions of the reviewer in our discussion.

Sincerely
V Stadlbauer